Author's response to reviews

Title: Plectin as a prognostic marker in non-metastatic oral squamous cell carcinoma

Authors:

Oddveig G Rikardsen (oddveig.rikardsen@unn.no)
Synnøve N Magnussen (synnove.magnussen@uit.no)
Gunbjørg Svineng (gunbjorg.svineng@uit.no)
Elin Hadler-Olsen (elin.hadler-olsen@uit.no)
Lars Uhlin-Hansen (lars.uhlin-hansen@unn.no)
Sonja E Steigen (sonja.steigen@unn.no)

Version: 4 Date: 12 August 2015

Author's response to reviews: see over
To Editor, BMC Oral Health

From Sonja E. Steigen, Department of Diagnostic Clinic – Clinical Pathology, University Hospital of North Norway, Tromsø, Norway and Department of Medical Biology - Tumor Biology Research Group, Faculty of Health Sciences, UiT The Arctic University of Norway, Tromsø, Norway.

Dear Editor,

Thank you for reviewing our manuscript MS: 1689717504163520 entitled: “Plectin as a prognostic marker in non-metastatic oral squamous cell carcinoma”, and giving us the opportunity to submit a revised version.

The manuscript has been revised according to the suggestions from Biomed Central Editorial:

1) The title in the Introduction section has been changed to “Background”.
2) We have included a Conclusion section after the Discussion

The reviewer had no further comments

Sincerely,

Sonja E. Steigen
MD, PhD, University Hospital of North Norway, Tromsø, Norway
Professor, UiT The Arctic University of Norway, Tromsø, Norway